Author’s response to reviews

Title: Challenges and Opportunities towards the road of Universal Health Coverage (UHC) in Nepal: a systematic review

Authors:

Chhabi Ranabhat (chhabir@gmail.com)
Kim Bae (kimcb@yonsei.ac.kr)
Ajanta Singh (ajanta2034@hotmail.com)
Dev Acharaya (drabmc@gmail.com)
Krishna Pathak (krishnapathak@gmail.com)
Basundhara Sharma (basundharasharma@gmail.com)
Shiva Mishra (shivarajmishra@gmail.com)

Version: 3 Date: 10 Jan 2019

Author’s response to reviews:

Editorial comments

Please include information in the title of the prisma chart and tables related to the subject => systematic review and the period covert.

Dear Editor-in-chief,

Thank you for your editorial comment and coordinating to review our manuscript. We appreciate your role from the bottom of heart.

As you suggested, we have included title of PRISMA chart and table related to subject adding "systematic review" and period of published in range (year).

Chhabi Ranabhat and authors team.